Author's response to reviews

Title: The SOSTA - net trial: a randomised, multi-centre controlled trial of the group-based social skills training SOSTA-FRA in children and adolescents with high functioning Autism Spectrum Disorder

Authors:

Christine M Freitag (C.Freitag@em.uni-frankfurt.de)
Hannah Cholemkery (hannah.cholemkery@kgu.de)
Leyla Elsuni (leyla.elsuni@kgu.de)
Anne K Kröger (anne.kroeger@kgu.de)
Stephan Bender (Stephan.Bender@uniklinikum-dresden.de)
Cornelia U Kunz (meinhard.kieser@imbi.uni-heidelberg.de)
Meinhard Kieser (meinhard.kieser@imbi.uni-heidelberg.de)

Version: 3 Date: 19 July 2012

Author's response to reviews:

Dear Editors,

In this revised version, two references were changed. The rest of the manuscript is unchanged.

Yours sincerely,

Christine Freitag